Author's response to reviews

Title: The limited usefulness of real-time 3-dimensional echocardiography in obtaining normal reference ranges for right ventricular volumes

Authors:

Erlend Aune (erlend.aune@siv.no)
Morten Baekkevar (morten.baekkevar@siv.no)
Olaf Rodevand (oroedev@frisurf.no)
Jan Erik Otterstad (jan.ek.otterstad@siv.no)

Version: 3 Date: 2 July 2009

Author's response to reviews:

Dear Editorial Production Team,

Attached you find the latest revision of our manuscript. According to your instructions we have renamed fig. 1a and 1b into 1 and 2, respectively. We have also made changes according to your "minor revisions".

Finally, we have proof-read the manuscript and made a few typographical corrections.

Kind regards,
Erlend Aune